

DOI: 10.5505/anatoljfm.2021.69133 Anatol J Family Med 2022;5(1):50–55

# Attitudes of Family Health Professionals on COVID-19 Vaccines and an Evaluation of Underlying Factors in Samandağ, Hatay, Turkey

💿 Duygu Kavuncuoğlu, 💿 Mehmet Mustafa Yıldırım

Department of Infectious Diseases, Samandağ Health Directorate, Hatay, Turkey

### ABSTRACT

**Objectives:** This study aimed to evaluate the attitudes of family healthcare providers on coronavirus disease 2019 (COVID-19) vaccines.

**Methods:** In this cross-sectional study, 67 (91.8%) of the total 73 family healthcare professionals in Samandağ, Hatay, Turkey, were included. A questionnaire comprising 17 questions, including questions on the sociode-mographic and occupational characteristics of the health professionals and their attitude toward COVID-19 vaccines, was used.

**Results:** In the study, 67 healthcare professionals were included. The median age was 40.0 (26.0–62.0) years, and 40 (59.7%) of the participants were females. Of the participants, 19 (28.4%) had been diagnosed with at least one chronic disease, and 14 (20.9%) were living together with an individual over 65 years of age. Of the participants, 24 (35.8%) stated their willingness to get vaccinated if a COVID-19 vaccine gets approved in Turkey, and 29 (43.3%) of the participants did not decide. Fifteen (22.4%) professionals stated lack of comprehensive information on the vaccine as the main reason for their refusal to get vaccinated for COVID-19. The rate of physicians stating that they would get vaccinated was 30 (69.8%) and the rate of midwives/nurses was 13 (30.2%) (p=0.001).

**Conclusion:** Vaccine hesitancy and refusal against COVID-19 are very important issues among family healthcare professionals. It is extremely important to organize in-service training and to ensure that all professionals have access to knowledge on the COVID-19 vaccines.

Keywords: Attitude, COVID-19, vaccines

## **INTRODUCTION**

Vaccination is one of the basic building blocks of preventive medicine to ensure the protection and improvement of health globally.<sup>[1,2]</sup> The coronavirus pandemic has affected millions of people worldwide and has led to considerable effort for creating a safe, effective vaccine to prevent infection or minimize symptoms and reduce fatalities. In the long term, the development of vaccines against coronavirus disease 2019 (COVID-19) and providing global access to them are the main requirement to end this pandemic. However, the success of this strategy depends on the willingness of the people to get vaccinated.

With the availability of the very first vaccines against COVID-19, the issues of vaccine hesitancy and refusal have come to the attention of public health professionals and policymakers.<sup>[3,4]</sup> SAGE



Please cite this article as: Kavuncuoğlu D, Yıldırım MM. Attitudes of Family Health Professionals on COVID-19 Vaccines and an Evaluation of Underlying Factors in Samandağ, Hatay, Turkey. Anatol J Family Med 2022;5(1):50–55.

#### Address for correspondence: Dr. Duygu Kavuncuoğlu.

Department of Infectious Diseases, Samandağ Health Directorate, Hatay, Turkey

Phone: +90 530 885 02 60

E-mail: duygu\_koylu@hotmail.com

Received Date: 31.03.2021 Accepted Date: 30.07.2021 Published online: 28.04.2022

©Copyright 2022 by Anatolian Journal of Family Medicine -Available online at www.anatoljfm.org





Working Group on Vaccine Hesitancy stated, "vaccine hesitation is delaying or refusing to accept vaccination despite the availability of vaccination services."Vaccine refusal is defined as refusing to be vaccinated without any medical reason. Determinants affecting vaccine hesitancy and refusal are evaluated in three groups: contextual factors (impacts of historical, sociocultural, environmental, health system/ institutional, economic, or political factors), individual and group factors (impacts of personal or social/peer environment perceptions on vaccines), factors related to vaccine itself and vaccination-related effects (the reliability of vaccine or vaccination equipment, vaccination program, costs, knowledge level, and/or attitude of health professionals). Vaccine hesitancy and refusal will create difficulties in immunizing a sufficient percentage of the population to control the epidemic. They also pose threat not only to the individuals who are not vaccinated but also to the health of the whole society. Vaccine hesitancy and refusal were acknowledged by the World Health Organization (WHO) as "one of the top ten threats to global health" in 2019.<sup>[5]</sup> As this is a complex phenomenon involving different factors, it is important to understand, take necessary precautions, and make interventions during the decision-making process for vaccination policies.

As stated in the family medicine practice regulations, immunization services should be carried out by family physicians in Turkey.<sup>[6,7]</sup> In addition, family physicians and other family healthcare professionals have crucial roles in providing community education, answering people's questions, and solving problems related to health services as well as being role models for proper healthcare decisions. Therefore, determining the attitudes of family physicians and other family healthcare professionals will be important before implementing vaccination policies.

In the review conducted on the vaccination rates in healthcare professionals in Turkey in 2017, it was seen that the rate of having the influenza vaccine varied between 12.3% and 35.3%.<sup>[8]</sup> There are a limited number of studies on hepatitis B vaccination rates of healthcare professionals, and the rate was found to be around 85%–90% in the studies. There are insufficient data on measles–mumps–rubella and varicella vaccine rates, and it has been estimated that the immunity rate is over 90% for these diseases. In another study conducted in 2019, it was found that 10.5% of healthcare professionals did not want themselves and their children to get vaccinated.<sup>[9]</sup>

This study aimed to evaluate the attitudes of family health center professionals on the COVID-19 vaccines.

### **METHOD**

This cross-sectional study was conducted on January 11– 12, 2020, in Family Health Centers in the Samandağ district of Hatay, Turkey. There was a total of 73 health workers, including 39 (53.4%) family physicians and 34 (46.6%) other family health professionals, working in the district of Samandağ. It was aimed to reach all healthcare professionals working in Family Health Centers to participate in the study. The sample size was not calculated in this study. Of the 73 physicians and health professionals, 67 (91.8%) of them, who were contacted by the authors, agreed to be included in the study.

A questionnaire was used as a data collection tool. It consisted of 17 questions on the sociodemographic and occupational characteristics of the participants and their attitudes toward COVID-19 vaccines.

The analysis of the data was done using the IBM Statistical Package for the Social Sciences (SPSS) 25.0 (SPSS, Inc., Chicago, IL, USA) program. Normality tests were carried out using the Kolmogorov–Smirnov and Shapiro-Wilk tests. Frequency, percentage, median, maximum, and minimum were used for descriptive statistics. The Chi-squared test, Fisher's exact test, and the Mann–Whitney U tests were used in group comparisons. The statistical significance threshold was considered at p<0.05.

### RESULTS

In the study, 67 healthcare professionals were included. According to their professionals, 16 (23.9%) midwives, 21 (31.3%) nurses, and 30 (44.8%) physicians participated in the study. The median age of the healthcare professionals was 40.0 (26.0–62.0) years. Female participants were 40 (59.7%), 19 (28.4%) had at least one diagnosed chronic disease, and 14 (20.9%) were living together with and/or taking care of an individual over 65 years of age in their households. The sociodemographic characteristics of the participants are summarized in Table 1.

The median professional experience of the participants was 15.0 (2.0–41.0) years. Of the participants, 42 (62.7%) of them stated that they had Hepatitis B vaccine while working as a health professional. The conjugated pneumococcus vaccination rate was 25 (37.3%), and the seasonal influenza vaccination rate was 18 (26.9%). The vaccination history of the participants is summarized in Table 2.

Of the participants, 24 (35.8%) stated their willingness to get vaccinated if a COVID-19 vaccine gets approved in Turkey, and 29 (43.3%) of the participants did not decide. In

**Table 1.** Sociodemographic characteristics of theparticipants

|  | n (%)     |
|--|-----------|
| Gender   |           |
| Female   | 40 (59.7) |
| Male   | 27 (40.3) |
| Education status   |           |
| Associate / college  | 9 (13.4)  |
| University   | 50 (74.7) |
| Expert/PhD   | 8 (11.9)  |
| Profession   |           |
| Midwife  | 16 (23.9) |
| Nurse  | 21 (31.3) |
| Physician  | 30 (44.8) |
| Chronic disease  |           |
| Yes  | 19 (28.4) |
| No   | 48 (71.6) |
| Having a child   |           |
| Yes  | 56 (83.6) |
| No   | 11 (16.4) |
| Individuals ( $\geq$ 65 years) requiring assistance in the hou | usehold   |
| Yes  | 14 (20.9) |
| No   | 53 (79.1) |

the study, 14 (20.9%) refused to get vaccinated, and further inquiries about their nonacceptance were made in this group. Vaccine refusal reasons of the participants are shown in Figure 1.

The median age of professionals who would recommend COVID-19 vaccines to all patients was 44.0 (26.0–62.0) years and that of those who would not recommend was 39.0 (26.0–59.0) years (p=0.186). In addition, the median dura-

| Table 2. Vaccination history of the participants |           |  |
|--|-----------|--|
|  | n (%)     |  |
| Tetanus  | 45 (67.2) |  |
| Hepatitis B                                      | 42 (62.7) |  |
| Conjugate pneumococcal vaccine                   | 25 (37.3) |  |
| Influenza  | 18 (26.9) |  |
| Measles  | 9 (13.4)  |  |
| H1N1 vaccine                                     | 6 (9.0)   |  |
| Rubella  | 5 (7.5)   |  |
| Mumps  | 5 (7.5)   |  |
| Varicella  | 2 (3.0)   |  |
| Hepatitis A                                      | 2 (3.0)   |  |



Figure 1. Vaccine refusal reasons of participants.

tion of the profession of professionals who would recommend COVID-19 vaccines to all patients was 20.5 (2.0–40.0) years and that of those who would not recommend was 15.0 (2.0–41.0) years (p=0.326). The sociodemographic characteristics of the participants according to their status of being vaccinated against COVID-19 are summarized in Table 3.

Of the participants, 23 (34.3%) of them stated that they would recommend COVID-19 vaccines to all patients, and 37 (55.2%) stated that they would recommend it only to patients in the risk group. Of the family healthcare professionals who participated in the study, 21 (31.3%) stated that they would recommend COVID-19 vaccination to all

# **Table 3.** Sociodemographic characteristics of theparticipants according to their status of being vaccinatedagainst COVID-19

|   | Yes (n=24) | No/undecided (n=43) | р                  |  |
|---|------------|---------------------|--------------------|--|
| Gender  |            |                     |                    |  |
| Female  | 11 (45.8)  | 29 (67.4)           | 0.084*             |  |
| Male  | 13 (54.2)  | 14 (32.6)           |                    |  |
| Chronic disease   |            |                     |                    |  |
| Yes   | 10 (41.7)  | 9 (20.9)            | 0.071*             |  |
| No  | 14 (58.3)  | 34 (79.1)           |                    |  |
| Having a child  |            |                     |                    |  |
| Yes   | 18 (75.0)  | 38 (88.4)           | 0.182 <sup>+</sup> |  |
| No  | 6 (25.0)   | 5 (11.6)            |                    |  |
| Individuals ( $\geq$ 65 years) requiring assistance in the household    |            |                     |                    |  |
| Yes   | 4 (16.7)   | 10 (23.3)           | 0.525*             |  |
| No  | 20 (83.3)  | 33 (76.7)           |                    |  |
| Profession  |            |                     |                    |  |
| Midwife/nurse   | 7 (29.2)   | 30 (69.8)           | 0.001*             |  |
| Physician   | 17 (70.8)  | 13 (30.2)           |                    |  |
| Data are presented as n (%).<br>*Chi-square test, †Fisher's exact test. |            |                     |                    |  |

their relatives, and 37 (55.2%) stated that they would do so only to those in the risk group. When asked about the criteria for "high-risk group" regarding COVID-19, 42 (62.7%) participants stated "being over 65 years old," 40 (59.7%) stated "having a chronic disease," 39 (58.2%) stated "being healthcare workers," and 35 (52.2%) stated "being immunosuppressed" as the main criteria for inclusion in the "highrisk" classification. Of the healthcare professionals, 47 (70.1%) stated that they followed updates on the COVID-19 disease through the Internet, 11 (16.4%) through journals and newspapers, and 38 (56.7%) through the communication networks of professional associations.

### DISCUSSION

Understanding the hesitations of healthcare professionals toward vaccines has important implications for the management of the COVID-19 pandemic. Healthcare professionals are at the forefront of combating the pandemic, and they routinely perform tasks involving the risk of contact with infectious agents. If they become infected with SARS-CoV-2, healthcare services would also be brought to a halt. In addition, healthcare professionals are the most direct and reliable sources of information for the public about vaccination.<sup>[10]</sup> WHO Vaccine Advisory Group also emphasized the role of healthcare professionals in building public trust in vaccines.<sup>[11]</sup> Moreover, it has been shown that healthcare workers whose attitudes toward vaccination are more positive tend to have higher rates of vaccination within the community for which they provide healthcare services.[12]

In this study, 20.9% of family physicians and other healthcare workers stated that they do not want to get vaccinated when an approved COVID-19 vaccine becomes available in their country, and 43.3% have undecided about getting vaccinated. In a study conducted with healthcare professionals in Malta, researchers found that 26% of the participants would not get the COVID-19 vaccine, and 23% did not decide.<sup>[13]</sup> In another study conducted with nurses in Hong Kong, it was reported that 63% would get the CO-VID-19 vaccine.<sup>[14]</sup>

Furthermore, other studies investigated the attitudes toward COVID-19 vaccines within the general population. For instance, in a study conducted in the United States, 10.8% of the adult participants stated that they would not get the COVID-19 vaccine, and 31.6% did not decide.<sup>[15]</sup> In a study involving approximately 20000 adults in 27 countries, 74% of the adults stated that they plan to have the COVID-19 vaccine, and the rates varied between countries. Willingness to be vaccinated was highest in China (97%), Brazil (88%), and Australia (88%). The lowest rates were found in Russia (54%), Poland (56%), and Hungary (56%).<sup>[16]</sup> In another study conducted with university students, there was no difference between students studying in the field of health services and other university students.<sup>[17]</sup> These studies showed that there were significant differences in the attitudes toward the COVID-19 vaccination depending on the place where the study was conducted and the target population.

However, it is clear that the issue of hesitancy/refusal toward the COVID-19 vaccine is a significant problem in all studied groups. In the limited number of studies conducted among healthcare professionals, it appears that the hesitancy/refusal rates of the COVID-19 vaccines were similar to that of other populations.

In this study, the most common reason for vaccine rejection among healthcare workers was the lack of comprehensive knowledge about the vaccine (22.4% of the participants hesitating/refusing to be vaccinated). In the Maltase study, 34% of the participants also stated that they had insufficient knowledge about the COVID-19 vaccine, and 51% were afraid of its long-term side effects.<sup>[13]</sup> Studies performed in different countries highlight the additional training needs of healthcare professionals on vaccines, providing adequate knowledge and the confidence to provide guidance on immunization.<sup>[18-21]</sup> In a systematic review, Herzog et al. showed that the attitudes and behaviors of health professionals who were given adequate training were effective in increasing the vaccination rates in the community.[22] In-service training on COVID-19 vaccines and immunity-related issues for the family health professionals, who have the most important role in vaccination services, will contribute to dispelling refusal or hesitancy toward vaccines in the general public.

In this study, it was found that the rate of physicians who were willing to get the COVID-19 vaccine was higher than that of other family healthcare professionals who were not willing to get the vaccine. These results were in line with the findings of a similar study conducted in Malta where the rate of physicians interested to be vaccinated for CO-VID-19 was higher.<sup>[13]</sup> This study also found a relationship between education level and the knowledge on immunization, as well as between the level of knowledge and attitudes toward vaccination.<sup>[23]</sup> However, in this study, the information level about the COVID vaccine was not investigated, and comprehensive studies are needed to explain the difference between nurses and physicians.

It was found that 34.4% of the participants would recommend COVID-19 vaccine to all patients, 31% to their relatives, and 55.2% to only patients in the risk group and their relatives in the risk group in this study.<sup>[24]</sup> In research conducted with physicians in Turkey, 90.7% of the participants reported that they advised and supported their relatives and family members about immunization. In a Finnish study conducted with healthcare professionals, 86% of the participants stated that they would recommend the people for whom they provide healthcare to get vaccinated.<sup>[25]</sup> These results demonstrate a lower rate of recommendation for COVID-19 vaccines by healthcare professionals when compared with the rates reported in previous studies.

In this study, 70.1% of healthcare professionals reported following updates on the COVID-19 pandemic through the Internet and 56.7% through the communication networks of professional associations. In this newly emerging and rapidly developing process of information flow, it is expected that the internet will be the most important source of information. However, the importance of the communication networks of professional associations during this period should not be ignored.

This study was conducted with family physicians and other family healthcare professionals who are directly responsible for the provision of vaccination services. It is important to determine the family medicine professionals' approach and attitudes toward vaccines not only in relation to service delivery but also due to their contribution to community education. However, as this study was conducted with a limited number of staff working for family medicine services in the Samandağ district, the results cannot be generalized.

### CONCLUSION

The hesitancy/refusal against COVID-19 vaccines is a very important issue among family healthcare professionals. In this study, it was observed that the nurses/midwives were more hesitant about vaccination than the doctors. Insufficient information about the vaccines was the most important factor underlying vaccine hesitancy/refusal. Inevitably, the vaccine hesitancy and/or refusal of the health professionals would have larger consequences for public health, leading to inadequate counseling and lack of guidance for the general public. It is important to organize in-service training to all healthcare professionals and to ensure that anyone working in the family health center should have the necessary and up-to-date information about the CO-VID-19 vaccines.

### Disclosures

Peer-review: Externally peer-reviewed.

**Conflict of Interest:** There is no conflict of interest between authors for this study.

**Ethics Committee Approval:** Ethical approval was obtained from the Mustafa Kemal University ethics committee (Approval date: Nov 12, 2020, and Approval number: 27). In addition, the approval of the Ministry of Health for research on the topic of CO-VID-19 and administrative permission from the Health Directorate were obtained prior to the study. Written and oral consent of the participants was obtained before the study.

Authorship Contributions: Concept – D.K.; Design – D.K.; Supervision – D.K.; Materials – D.K.; Data collection and/or processing – D.K., M.M.Y.; Analysis and/or interpretation – D.K., M.M.Y.; Literature search – D.K., M.M.Y.; Writing – D.K., M.M.Y.; Critical review – D.K.

### REFERENCES

- Dror AA, Eisenbach N, Taiber S, Morozov NG, Mizrachi M, Zigron A, et al. Vaccine hesitancy: the next challenge in the fight against COVID-19. Eur J Epidemiol 2020;35:775–9. [CrossRef]
- Yamey G, Schäferhoff M, Hatchett R, Pate M, Zhao F, McDade KK. Ensuring global access to COVID-19 vaccines. Lancet 2020;395(10234):1405–6. [CrossRef]
- M. Schuster, J. Eskola, P. Duclos, the SAGE working group on vaccine hesitancy. Review of vaccine hesitancy: rationale, remit and methods. Vaccine 2015;33(34):4157–60. [CrossRef]
- HJ Larson, Jarrett C, Eckersberger E, Smith DMD, Paterson P. Understanding vaccine hesitancy around vaccines and vaccination from a global perspective: a systematic review of published literature, 2007–2012. Vaccine 2014;32(19):2150–9.
- WHO. Ten threats to global health in 2019. Available at: https://www.who.int/news-room/spotlight/ten-threats-toglobal-health-in-2019. Accessed Dec 10, 2020.
- Aile hekimligi uygulama yonetmeligi; 25 Ocak 2013; sayı: 28539. Available at: https://www.resmigazete.gov.tr/eskiler/2013/01/20130125-26.htm. Accessed Dec 10, 2020.
- Karaoglu N. Tip egitiminde rol modellik ve aile hekimligi için onemi. TJFMPC 2012;6(2):30–5.
- Özisik L, Tanriover MD, Altınel S, Unal S. Vaccinating healthcare workers: Level of implementation, barriers and proposal for evidence-based policies in Turkey. Hum Vaccines Immunother 2017;13(5):1198–206. [CrossRef]
- Arıcan MD. Sağlık çalışanları arasında aşılanmaya genel bakış, aşı kabulü ve reddini etkileyen faktörler. Uzmanlık tezi. İzmir: Sağlık Bilimleri Üniversitesi İzmir Tepecik Eğitim ve Araştırma Hastanesi Aile Hekimliği Kliniği; 2019.
- European Centre for Disease Prevention and Control. Vaccine hesitancy among healthcare workers and their patients in Europe – a qualitative study. Stockholm: ECDC; 2015. Available at: https://www.ecdc.europa.eu/sites/default/files/media/en/

publications/Publications/vaccine-hesitancy-among-healthcare-workers.pdf. Accessed Mar 1, 2022.

- 11. WHO. Improving vaccination demand and addressing hesitancy. Available at: http://awareness.who.int/immunization/ programmes\_systems/vaccine\_hesitancy/en/ Accessed Dec 12, 2020.
- Paterson P, Meurice F, Stanberry LR, Glismann S, Rosenthal SL, Larson HJ. Vaccine hesitancy and healthcare providers. Vaccine 2016;34(52):6700–6. [CrossRef]
- 13. Grech V, Gauci C, Agius S. Vaccine hesitancy among Maltese healthcare workers toward influenza and novel COVID-19 vaccination. Early Hum Dev 2020:105213.
- 14. Kwok KO, Li KK, Wei WI, Tang A, Wong SYS, Lee SS. Influenza vaccine uptake, COVID-19 vaccination intention and vaccine hesitancy among nurses: A survey. Int J Nurs Stud 2021;114(2021):103854. [CrossRef]
- Peretti-Watel P, Seror V, Cortaredona S, Launay O, Raude J, Verger P, et al. A future vaccination campaign against CO-VID-19 at risk of vaccine hesitancy and politicisation. Lancet Infect Dis 2020;20(7):769–70. [CrossRef]
- Dubé E, Vivion M, MacDonald NE. Vaccine hesitancy, vaccine refusal and the anti-vaccine movement: influence, impact and implications. Expert Rev Vaccines 2015;14(1):99–117.
- Barello S, Nania T, Dellafiore F, Graffigna G, Caruso R. Vaccine hesitancy' among university students in Italy during the CO-VID-19 pandemic. Eur J Epidemiol 2020;35:781–3. [CrossRef]
- Ravlija J, Ivankovic A. Importance of health workers' communication in immunisation programmes. Healthmed 2012;6(2):672–7.

- Ishola Jr DA, Permalloo N, Cordery RJ, Anderson SR. Midwives' influenza vaccine uptake and their views on vaccination of pregnant women. J Public Health Med 2013;35(4):570–7.
- 20. Picchio CA, Carrasco MG, Sagué-Vilavella M, Rius C. Knowledge, attitudes and beliefs about vaccination in primary healthcare workers involved in the administration of systematic childhood vaccines, Barcelona, 2016/17. Euro Surveill 2019;24(6):1800117. [CrossRef]
- 21. Paoli S, Lorini C, Puggelli F, Sala A, Grazzini M, Paolini D, et al. Assessing vaccine hesitancy among healthcare workers: a cross-sectional study at an Italian paediatric hospital and the development of a healthcare worker's vaccination compliance index. Vaccines 2019;7(4):201. [CrossRef]
- 22. Herzog R, Álvarez-Pasquin MJ, Díaz C, Barrio JLD, Estrada JM, Gil A. Are healthcare workers' intentions to vaccinate related to their knowledge, beliefs and attitudes? A systematic review. BMC Public Health 2013;13:154. [CrossRef]
- 23. Bonville CA, Domachowske JB, Cibula DA, Suryadevara M. Immunization attitudes and practices among family medicine providers. Hum Vaccines Immunother 2017;13(11):2646–53.
- 24. Han-Yekdes D, Altunok A, Eskiocak M, Marangoz B. Bir universite hastanesindeki hekimlerin bagisiklamayla ilgili tutumlari. Klimik Dergisi 2020;33(3):255–9.
- 25. Karlsson LC, Lewandowsky S, Antfolk J, Salo P, Lindfelt M, Oksanen T, et al. The association between vaccination confidence, vaccination behavior, and willingness to recommend vaccines among Finnish healthcare workers. Plos One 2019;14(10):e0224330. [CrossRef]